\documentclass[gbc]{agutex}
\usepackage{longtable, lscape}

\begin{document}
{\small
\begin{landscape}
\begin{longtable}{|p{0.7cm}|p{2cm}|p{3.5cm}|p{1.3cm}|c|p{10cm}|}
\caption*{Table S1: List of the data contributors to the DMS database, with their contributor number, sampling region, number of samples, dates, and references.} \\
\endfirsthead
\caption*{... Table S1: (continued) ...} \\ 
\endhead
\hline
CN & Contributor & Region & Samples & Start-End dates & Reference(s) \\
\hline
1 &	KettleD &  Atlantic &	 20 &	 1972-03-11 1972-04-03 & \cite{Lovelock1972}, \cite{Liss1997} \\
2 &	KettleD & 	Atlantic, Indian &	19 & 1977-03-15 1978-05-15 & \cite{Nguyen1978} \\
3 &	Andreae	&  Florida Strait & 40 & 1980-04-07 1980-04-16 & \cite{Andreae1983} \\
4 &	Andreae &  Atlantic &	231 &	1980-10-09 	1980-11-07 &	\cite{Barnard1982}, \cite{Andreae1984} \\
5 &	Andreae &  Bering Sea &	13 &	1981-05-04 	1981-05-24 &	\cite{Barnard1984} \\
6 &	Andreae & 	Sargasso Sea &	67 &	1981-09-18 	1981-09-28 & \cite{Andreae1984} \\
7 &	Andreae & 	Charlotte Harbour &	30 &	1981-10-31 	1981-10-31 &	\cite{Froelich1985} \\
8 &	Bates &		Pacific &	70 &	1982-05-08 	1982-05-21 &	\cite{Cline1983}, \cite{Bates1987}, \cite{Bates1997} \\
9 &	Andreae & 	Peru Shelf &	294 &	1982-06-23 	1982-08-06 &	\cite{Andreae1985}, \cite{Andreae1983} \\
10 & Bingemer &	Southern, Atlantic &	89 &	1983-01-04 	1983-01-19 & \cite{Bingemer1984}, \cite{Bingemer1987} \\
11 & Bates & 	Pacific &	24 &	1983-03-04 	1983-03-20 &	\cite{Bates1997}, \cite{Bates1987} \\
12 & Bingemer &	Southern, Atlantic &	79 &	1983-03-30 	1983-04-22 &	\cite{Bingemer1984}, \cite{Bingemer1987} \\
13 & Bates 	& 	Pacific &	267 &	1983-04-02 	1983-05-01 &	\cite{Bates1997}, \cite{Bates1987} \\
14 &  Bates & 	Pacific &	123 &	1983-05-13 	1983-05-22 &	\cite{Bates1987}, \cite{Bates1985}, \cite{Bates1990} \\
15 & Andreae &	Bahamas &	99 &	1983-11-06 	1983-11-22 &	\cite{Andreae1985} \\
16 & Bates &  	Pacific &	48 &	1984-02-15 	1984-02-23 &	\cite{Bates1987}, \cite{Bates1997}, \cite{Bates1985} \\
17 & Bates & 	Pacific &	7 &	1984-03-20 	1984-03-22 &	\cite{Bates1987},  \cite{Bates1997} \\
18 & Bates & 	Pacific &	53 & 1984-04-02 1984-04-28 &	\cite{Bates1987}, \cite{Bates1997} \\
19 & Andreae &	Atlantic &	97 & 1984-04-24  1984-05-11 & \cite{Andreae1985} \\
20 & KettleD &	English Channel &	27 &	1984-06-18 	1984-06-19 & \cite{Holligan1987} \\
21 &  Bates &	Pacific &	62 &	1984-08-28 	1984-09-27 &	\cite{Bates1985}, \cite{Bates1987}, \cite{Bates1990} \\
22 & Turner &	English Channel &	176 &	1985-01-06 	1985-01-19 & \cite{Turner1988}, \cite{Turner1989} \\
23 & Turner &	Oosterschelde &	64 &	1985-05-01 	1985-05-03 &	unpublished \\
24 & Bates & Pacific &	116 &	1985-05-14 	1985-06-09 &	\cite{Bates1987}, \cite{Bates1990} \\
25 & Bates & Pacific, Arctic &	211 &	1985-06-29 	1985-10-08 &	\cite{Bates1987}, \cite{Bates1990} \\
26 & Turner & around Britain &	186 &	1985-07-10 	1985-08-02 &	\cite{Turner1988} \\
27 & Andreae & Mid-Atlantic Bight &	224 &	1986-02-03 	1986-02-15 &	\cite{Iverson1989}, \cite{Berresheim1991} \\
28 & Berresheim & Southern &	141 &	1986-03-21 	1986-04-26 &	\cite{Berresheim1989}, \cite{Berresheim1987} \\
29 & Bates & Pacific &	53 &	1986-04-22 	1986-04-30 &	\cite{Bates1990} \\
30 & Andreae & Mid-Atlantic Bight &	114 &	1986-04-22 	1986-05-02 &	\cite{Iverson1989}, \cite{Berresheim1991} \\
31 & Turner & southern North Sea &	151 &	1986-05-01 	1986-05-13 &	\cite{Turner1989} \\
32 & Bates & Pacific & 27 &	1986-06-11 	1986-06-14 &	\cite{Bates1990} \\
33 & Andreae &	Mid-Atlantic Bight &	153 &	1986-09-02 	1986-09-13 & 	\cite{Iverson1989}, \cite{Berresheim1991} \\
34 & KettleD &  Baltic Sea & 24 &	1987-03-28 	1988-06-09 &	\cite{Leck1990} \\
35 & KettleD &		Indian & 	23 &	1987-04-01 	1988-02-26 &	\cite{Nguyen1990} \\
36 & Bingemer &	 Atlantic &	98 &	1987-03-23 	1987-04-17 &	\cite{Burgermeister1990} \\
37 & Nguyen, Putaud, Mihalopoulos & Indian Ocean, Mediterranean &	66 &	1986-04-03 	1987-07-25 &	\cite{Mihalopoulos1989}, \cite{Mihalopoulos1992} \\
38 & Turner & 	 northern North Sea &	162 &	1987-04-23 	1987-05-10 & \cite{Turner1989} \\
39 & KettleD &	Antarctica &	14 &	1987-05-12 	1988-01-21 &	\cite{Gibson1988} \\
40 & Bates &	Pacific &	55 &	1987-05-14 	1987-05-21 &	\cite{Bates1990} \\
41 & Turner & 	NE Atlantic &	158 &	1987-06-09 	1987-07-01 &	\cite{Turner1989} \\
42 & Bates & Indian, Pacific &	100 &	1987-06-16 	1987-07-24 &	unpublished \\
43 & KettleD &		Baltic Sea &	14 &	1987-09-01 	1987-09-01 &	\cite{Leck1990} \\
44 & KettleD &		Pacific &	21 &	1988-01-26 	1988-03-10 &	\cite{Uchida1992} \\
45 & KettleD &	Indian &	22 &	1989-02-15 	1990-11-15 &	\cite{Nguyen1992} \\
46 & Bates & 	Pacific &	70 &	1988-04-08 	1988-05-05 &	\cite{Quinn1990}, \cite{Bates1997} \\
47 & Turner & North Sea &	55 &	1988-04-08 	1988-04-23 &	unpublished \\
48 & Leck &	 Baltic Sea &	24 &	1988-07-12 	1988-07-12 &	\cite{Leck1991} \\
49 & KettleD &	 Pacific &	29 &	1988-07-08 	1988-07-30 &	\cite{Watanabe1995b} \\
50 & Leck & Baltic Sea &	35 &	1988-07-19 	1988-07-20 &	\cite{Leck1991} \\
51 & Leck &	 North Sea &	30 &	1988-07-26 	1988-07-26 &	\cite{Leck1991} \\
52 & KettleD &	 Pacific &	37 &	1988-08-05 	1988-08-23 &	\cite{Watanabe1995b} \\
53 & Bingemer &	 Atlantic &	60 &	1988-09-17 	1988-10-05 &	\cite{Staubes1993b}, \cite{Staubes-Diederich1992}, \cite{Staubes1993a} \\
54 & McTaggart & Southern &	44 &	1988-11-06 	1989-01-24 &	\cite{McTaggart1992} \\
55 & Turner &  North Sea &	798 &	1989-01-29 	1989-10-02 &	\cite{Turner1996a}, \cite{Liss1993} \\
56 & Bates & Pacific &	108 &	1989-02-17 	1989-04-20 &	\cite{Bates1992}, \cite{Bates1997} \\
57 & KettleD &	Atlantic &	83 &	1989-03-14 	1989-04-03 &	\cite{Tanzer1992} \\
58 & Matrai & Sargasso Sea &	39 &	1989-04-22 	1989-04-25 & \cite{Matrai1996} \\
59 & Bates & Pacific &	62 &	1989-05-31 	1989-06-09 &	unpublished \\
60 & Staubes &	Atlantic &	102 &	1989-08-06 	1989-09-01 &	\cite{Staubes-Diederich1992}, \cite{Staubes1993a}, \cite{Staubes1993b} \\
61 & Matrai & 	near New Jersey &	9 &	1989-08-24 	1989-09-03 &	\cite{Matrai1993a} \\
62 & Matrai & 	New England &	12 &	1989-08-21 	1989-08-23 &	\cite{Matrai1993a} \\
63 & Helas, Schebeske, Andreae &  Southern &	62 &	1990-01-29 	1990-03-06 &	unpublished \\
64 & Bates &  Pacific &	744 &	1990-02-20 	1990-03-09 &	\cite{Bates1993}, \cite{Bates1997} \\
65 & Bates &  Pacific &	136 &	1990-04-09 	1990-04-26 &	unpublished \\
66 & KettleD & Mediterranean Sea & 17 &	1991-05-16 	1991-05-22 &	\cite{Belviso1993} \\
67 & Matrai &  Gulf of Maine &	8 &	1990-07-08 	1990-07-13 &	\cite{Matrai1993} \\
68 & Keller &  Gulf of Maine &	10 &	1990-07-08 	1990-07-13 &	\cite{Matrai1993} \\
69 & Staubes &	Greenland Sea &	85 &	1990-07-12 	1990-08-09 &	\cite{Staubes1993a}, \cite{Staubes1993b}, \cite{Staubes-Diederich1992} \\
70 & Keller &  Gulf of Maine & 	27 &	1990-08-07 	1991-10-07 &	unpublished \\
71 & Staubes &	Southern, Atlantic &	210 &	1990-10-22 	1990-12-22 &	\cite{Staubes1993a}, \cite{Staubes1993b}, \cite{Staubes-Diederich1992} \\
72 & Andreae & Atlantic &	342 &	1991-02-12 	1991-03-20 &	\cite{Andreae1994} \\
74 & Bates & Pacific &	616 &	1991-04-16 	1991-05-01 &	\cite{Bates1994} \\ 
75 & Keller &  Gulf of Maine &	38 &	1991-04-23 	1991-04-29 &	unpublished \\
76 & Turner & Atlantic &	152 &	1991-06-14 	1991-06-27 &	\cite{Holligan1993} \\
77 & Keller & Gulf of Maine &	39 &	1991-07-06 	1991-07-14 &	unpublished \\
78 & Leck & Arctic &	303 &	1991-08-04 	1991-10-06 &	\cite{Leck1996} \\
79 & Nguyen, Putaud, Mihalopoulos & Atlantic & 	110 &	1991-09-18 	1991-10-23 &	\cite{Putaud1993b}, \cite{Putaud1993a} \\
80 & KettleD &	North Sea &	27 	& 1991-11-14 	1993-06-11 &	\cite{Kwint1996} \\
81 & Rapsomanikis & Southern &	52 &	1991-12-11 	1991-12-28 &	\cite{Kirst1993} \\
82 & KettleD &	Sargasso Sea & 	44 &	1992-01-29 	1993-10-25 &	\cite{Siegel1996} \\
83 & DiTullio &	Ross Sea &	30 &	1992-02-07 	1992-02-26 &	\cite{DiTullio1993}, \cite{DiTullio1995} \\
84 & Bates & Pacific &	952 &	1992-02-22 	1992-03-20 	& \cite{Kieber1996}, \cite{Yvon1996} \\
85 & Kiene & Pacific &	26 &	1992-02-24 	1992-03-08 & unpublished \\
86 & Keller & Gulf of Maine &	8 &	1992-04-02 	1992-04-10 &	unpublished \\
87 & Andreae & Atlantic &	89 &	1993-04-16 	1993-05-03 &	\cite{Pfannkuche1993} \\
88 & Boniforti & Mediterranean Sea 	& 78 &	1992-04-28 	1994-10-05 &	\cite{Boniforti1993}, unpublished \\
89 & Nguyen, Putaud, Mihalopoulos &	Atlantic &	70 &	1992-06-02 	1992-06-21 &	\cite{Putaud1996} \\
90 & KettleD &	 Atlantic &	76 &	1992-06-12 	1992-06-20 &	\cite{Blomquist1996} \\
91 & Keller & Gulf of Maine &	2 &	1992-07-12 	1992-07-14 &	unpublished \\
92 & KettleD &	Atlantic Ocean &	26 &	1992-09-20 	1992-10-19 &	\cite{Groene1995}, unpublished \\
93 & Curran, Jones & Great Barrier Reef &	12 &	1992-09-19 	1992-09-21 &	unpublished \\
94 & Curran, Jones & Tasman Sea &	18 &	1992-09-22 	1992-10-04 &	unpublished \\
95 & Turner & Southern Ocean &	125 &	1992-10-29 	1992-11-28 &	\cite{Turner1995} \\
96 & KettleD &	Southern Ocean &	39 &	1992-11-24 	1992-11-28 &	\cite{Turner1995} \\
97 & Turner & Southern Ocean &	101 &	1993-02-10 	1993-03-14 &	\cite{Chuck2005} \\
98 & KettleD & East China Sea &	50 &	1993-02-28 	1994-08-15 &	\cite{Uzuka1996} \\
99 & Bates & Pacific &	1308 &	1993-03-21 	1993-05-07 &	\cite{Bates1997} \\
100 & Matrai &	Barents Sea &	18 &	1993-05-13 	1993-05-29 &	\cite{Matrai1997} \\
101 & KettleD &	Jiaozhou Bay, China &	34 &	1993-01-15 	1994-09-15 &	\cite{Hu1997} \\
102 & Simo & Mediterranean Sea &	53 &	1993-06-01 	1994-07-21 &	\cite{Simo1997}, \cite{Simo1995} \\
103 & Rapsomanikis & Aegean Sea &	55 &	1993-07-04 	1993-07-15 &	unpublished \\
104 & Levasseur & Gulf of St. Lawrence &	64 &	1993-08-03 	1993-08-09 &	\cite{Cantin1996} \\
105 & KettleD & New Zealand &	37 &	1993-08-31 	1993-11-13 &	\cite{Lee1996} \\
106 & KettleD & Gulf Mexico &	38 &	1993-09-17 	1994-12-14 &	\cite{Kiene1996} \\
107 & Yang & East China Sea &	14 &	1993-10-15 	1994-10-15 &	\cite{Yang1996} \\
108 & Bingemer & Southern &	215 &	1993-10-21 	1994-03-21 &	unpublished \\
109 & Turner & Pacific &	9 &	1993-10-22 	1993-10-28 &	\cite{Turner1996b} \\
110 & Turner & Pacific &	9 &	1993-11-09 	1993-11-19 &	\cite{Hatton1998} \\
111 & Yang & South China Sea &	19 &	1993-11-15 	1993-11-15 &	\cite{Yang1999} \\
112 & Bates &	Pacific &	1172 &	1993-11-23 	1994-01-06 &	\cite{Bates1997} \\
113 & Berresheim &	Southern &	15 &	1994-01-20 	1994-02-22 &	\cite{Berresheim1998} \\
114 & Andreae, Schebeske &	Atlantic &	41 &	1994-04-08 	1994-05-06 & unpublished \\
115 & Sharma &	Atlantic, Arctic, Pacific &	43 &	1994-07-18 	1994-08-15 &	\cite{Sharma1999} \\
116 & Uher, Schebeske, Rapsomanikis, Andreae &	Atlantic &	208 &	1994-09-08 	1994-09-19 &	\cite{Uher1995} \\
117 & Curran, Jones & Southern &	56 &	1994-09-01 	1994-10-18 &	\cite{Curran1998} \\
118 & KettleD & Indian &	19 &	1994-09-09 	1994-09-11 &	\cite{Hatton1996}, \cite{Hatton1999} \\
119 & Amouroux, Andreae & Pacific &	89 &	1994-09-08 	1994-09-16 &	unpublished \\
120 & Curran, Jones & Southern &	21 & 	1994-12-20 	1995-01-31 &	\cite{Curran1998} \\
121 & KettleD & California coast &	10 &	1995-04-19 	1995-05-01 &	\cite{Ledyard1996} \\
122 & Turner & Pacific &	19 &	1995-05-25 	1995-06-07 &	\cite{Turner1996b} \\
123 & Keller & Gulf of Maine &	3 &	1995-06-17 	1995-06-21 	& unpublished \\
124 & Uher, Schebeske, Rapsomanikis, Andreae &	Atlantic &	393 &	1995-07-15 	1995-07-28 &	\cite{Uher1997}, \cite{Uher1996} \\
125 & Roberts, Amouroux, Andreae &	Black Sea &	9 &	1995-07-26 	1995-07-31 	& \cite{Amouroux2002} \\
126 & Andreae, Schebeske &	Atlantic &	49 &	1995-08-02 	1995-08-12 &	unpublished \\
128 & Bates & Pacific,Southern &	1206 &	1995-10-21 	1995-12-12 &	\cite{Bates1997}, \cite{Bates1998} \\
129 & Curran, Jones & Southern &	28 &	1995-11-18 	1995-12-05 &	\cite{Curran1998} \\
130 & Bates &	Pacific &	1068 &	1996-03-15 	1996-04-12 &	\cite{Bates1997} \\
131 & Leck 	& Arctic 	& 33 &	1996-07-15 	1996-08-06 	& unpublished \\
132 & DiTullio &	Ross Sea &	85 &	1996-12-16 	1997-01-06 &	unpublished \\
133 & Levasseur &	Labrador Sea &	37 &	1997-05-13 	1997-06-09 &	unpublished \\
134 & Kiene &	Gulf of Mexico &	13 &	1997-09-23 	1997-10-01 &	unpublished \\
135 & Belviso & Atlantic &	909 &	1996-10-09 	1996-11-05 &	\cite{Belviso2000} \\
136 & Sciare &	Indian &	463 &	1997-12-03 	1997-12-19 &	\cite{Sciare1999} \\
137 & Sciare &	Indian &	157 &	1998-08-19 	1998-08-30 &	unpublished \\
138 & Bates &	Atlantic &	806 &	1997-06-20 	1997-07-23 &	\cite{Bates1997} \\
139 & Bates &	Atlantic &	666 &	1999-01-15 	1999-02-08 & 	\cite{Bates2001} \\
140 & Bates &	Indian &	441 &	1999-02-11 	1999-03-29 &	unpublished \\
141 & Bates &	Pacific, Sea of Japan &	1535 &	2001-03-16 	2001-04-19 &	unpublished \\
142 & Johnson &	 Bering Sea - Gulf of Alaska &	879 &	2002-05-06 	2002-05-31 &	unpublished \\
143 & Johnson &	Bering Sea - Gulf of Alaska &	577 &	2002-06-07 	2002-06-22 &	unpublished \\
144 & Johnson &	US West Coast &	571 &	2002-11-07 	2002-11-22 &	unpublished \\
145 & Johnson &	Gulf of Alaska - Bering Sea &	2128 	& 2003-01-31 	2003-03-31 &	unpublished  \\
146 & Johnson &	Gulf of Alaska - Bering Sea &	2257 &	2003-04-30 	2003-06-24 &	unpublished \\
147 & Johnson &	Gulf of Alaska &	1949 &	2003-06-27 	2003-08-13 & unpublished \\
148 & Johnson &	Eastern Equatorial Pacific &	1057 &	2003-10-27 	2003-11-20 &	\cite{Huebert2004} \\
149 & Belviso &	North Atlantic, Mediterranean Sea &	552 &	1999-09-07 	1999-09-23 &	\cite{Belviso2003} \\
150 & Wong & Northeast Pacific &	20 &	1996-02-22 	1996-08-31 &	\cite{Wong2006}, \cite{Wong2005} \\
151 & Wong & Northeast Pacific &	17 &	1997-02-13 	1997-09-12 & \cite{Wong2006}, \cite{Wong2005} \\
152 & Wong & Northeast Pacific &	17 &	1998-02-20 	1998-09-04 &	\cite{Wong2005}, \cite{Wong2006} \\
153 & Wong & Northeast Pacific &	8 	& 1999-02-10 	1999-06-22 &	\cite{Wong2006}, \cite{Wong2005} \\
154 & Wong & Northeast Pacific &	11 	 & 2000-06-01 	2000-09-15 &	\cite{Wong2006}, \cite{Wong2005} \\
155 & Wong & Northeast Pacific &	8 	& 2001-06-18 	2001-06-25 &	\cite{Wong2006}, \cite{Wong2005} \\
156 & Kiene, Kieber & Southern Ocean &	58 &	2003-10-31 	2007-11-09 &	\cite{Kiene2007} \\
157 & Kiene, Kieber & Southern Ocean &	45 &	2004-12-19 	2004-12-27 &	\cite{Kiene2007} \\
158 & Kiene, Kieber & Southern Ocean &	39 &	2005-10-26 	2005-11-07 &	\cite{Kiene2007} \\
159 & Malin, Liss, Bell & Atlantic Ocean &	132 &	1997-09-17 	1997-10-16 & \cite{Bell2006} \\
160 & Malin, Liss, Bell &	Atlantic Ocean &	107 &	1999-09-19 	1999-10-12 & \cite{Bell2006} \\
161 & Malin, Liss, Bell &	Atlantic Ocean &	57 	& 2003-05-16 	2003-06-12 & \cite{Bell2006} \\
162 & Malin, Liss, Bell &	Atlantic Ocean &	26 	& 2003-10-02 	2003-10-11 & \cite{Bell2006} \\
163 & Malin, Liss, Bell &	Atlantic Ocean &	66 	& 2004-05-03 	2004-05-26 & \cite{Bell2006} \\
164 & Johnson &	Gulf of Maine &	1379 &	2004-07-09 	2004-08-12 &	unpublished \\
165 & Johnson &	Gulf of Mexico & 	431 &	2006-08-23 	2006-09-11 &	unpublished \\
166 & Johnson &	Eastern Equatorial Pacific &	572 	& 2006-10-14 	2006-10-26 &	unpublished \\
168 & Baker & Northeast Atlantic &	204 &	1997-05-12 	1997-05-29 &	\cite{Baker2000} \\
169 & Saltzman &	Equatorial, North Pacific Ocean &	214 &	2004-05-23 	2004-07-01 &	\cite{Marandino2007} \\
170 & Saltzman &	Southeastern Pacific Ocean 	& 82 &	2006-01-11 	2006-01-24 &	\cite{Marandino2009} \\
171 & Kiene, Kieber &	Southern Ocean &	21 &	2004-12-23 	2005-01-19 &	\cite{Valle2009}, \cite{Valle2007}, \cite{Valle2009a} \\
172 & Kiene, Kieber &	Southern Ocean &	11 &	2005-11-08 	2005-11-29 &	\cite{Valle2009}, \cite{Valle2007}, \cite{Valle2009a} \\
173 & Johnson & Atlantic Ocean &	105 &	2007-10-09 	2007-10-12 &	unpublished \\
174 & Johnson &	Eastern Tropical Pacific &	1070 &	2007-10-16 	2007-11-11 &	unpublished \\
175 & Johnson &	North Atlantic Ocean &	1579 &	2008-03-21 	2008-04-24 &	unpublished \\
176 & Turner &	East Atlantic Ocean &	403 &	1996-06-08 	1996-07-02 &	\cite{Locarnini1998} \\
177 & Turner &	East Atlantic Ocean &	210 &	1998-06-10 	1998-07-05 &	\cite{Jickells2008} \\
178 & Turner &	Southern Ocean, Atlantic Sector &	39 &	2000-11-03 	2000-11-06 &	\cite{Turner2004} \\
179 & Lee, de Mora, Marsden, Gratton &	North Water Polynya, Baffin Bay &	98 &	1998-04-12 	1998-06-24 &	\cite{Bouillon2002} \\
180 & Lee, DiTullio, Bruland &	Bering Sea &	80 &	2003-08-12 	2003-09-05 &	unpublished \\
181 & Lee, DiTullio, Hutchins, Leblanc, Wilhelm & 	North Atlantic Ocean &	76 &	2005-06-06 	2005-07-03 &	unpublished \\
182 & Lee, DiTullio, Dunbar, Smith &	Ross Sea &	188 &	2005-12-24 	2006-01-23 &	unpublished \\
183 & Lee, DiTullio, Dunbar, Smith &	Ross Sea &	110 &	2006-11-11 	2006-12-06 &	unpublished \\
184 & Lee, DiTullio &	New Zealand Sector, Southern Ocean &	212 &	2006-11-03 	2006-12-11 &	unpublished \\
185 & Lee, DiTullio &	South Atlantic Ocean &	57 &	2007-11-18 	2007-12-11 &	unpublished \\
186 & Simo 	& Mediterranean Sea &	37 &	2003-01-13 	2004-06-28 &	\cite{Vallina2007}, \cite{Vila-Costa2008} \\
187 & Simo 	& Mediterranean Sea &	11 &	2008-03-11 	2008-09-17 &	unpublished \\
188 & Simo 	& Arctic 	& 17 &	2007-07-01 	2007-07-22 & \cite{Gali2010} \\
189 & Smythe-Wright &	Mascarene Plateau, Indian Ocean &	57 &	2002-06-18 	2002-07-09 &	\cite{Smythe-Wright2005} \\
190 & Simo 	& Mediterranean Sea &	6 &	2007-09-18 	2007-09-25 &	unpublished \\
191 & Deal 	& Bering Sea 	& 6 &	2001-08-29 	2001-08-31 &	\cite{Deal2005} \\
192 & Scarratt &	NW Atlantic &	139 &	2003-04-25 	2003-05-15 &	\cite{Merzouk2008}, \cite{Lizotte2008} \\
193 & Scarratt &	NW Atlantic &	110 &	2003-10-13 	2003-10-30 &	unpublished \\
194 & Scarratt &	NW Atlantic &	255 &	1999-09-11 	1999-09-30 &	\cite{Scarratt2007} \\
195 & Franklin, Malin &	Mauritanian upwelling, Atlantic Ocean &	59 &	2006-07-13 	2006-08-02 &	unpublished \\
196 & Archer &	Southern Ocean &	2984 &	2008-03-03 	2008-04-07 &	unpublished \\
197 & Kasamatsu &	Southern Ocean (Indian sector) &	19 &	2005-01-03 	2005-01-20 &	unpublished \\
198 & Kasamatsu &	Southern Ocean (Indian sector) &	72 &	2006-01-06 	2006-01-26 &	unpublished \\
199 & Watanabe, Kasamatsu &	Okhotsk Sea &	27 &	2000-06-13 	2000-06-28 &	unpublished \\
200 & Kasamatsu, Odate, Watanbe & Southern Ocean (Indian sector) &	10 &	2004-01-31 	2004-02-10 &	unpublished \\
201 & Kasamatsu &	Western North Pacific &	60 &	2001-06-07 	2001-07-07 &	unpublished \\
202 & Kasamatsu &	Southern Ocean &	7 &	2003-01-28 	2003-03-06 &	\cite{Kasamatsu2004}, \cite{Kasamatsu2005}, \cite{Preunkert2007} \\
203 & Kasamatsu &	Central Southern Pacific, Southern Ocean &	46 &	2001-12-08 	2002-01-17 &	\cite{Kasamatsu2004}, \cite{Kasamatsu2005}, \cite{Preunkert2007} \\
204 & Kasamatsu &	Southern Ocean (Indian sector) &	53 &	2007-12-27 	2008-01-10 &	unpublished \\
205 & Kasamatsu &	Southern Ocean &	33 & 	2003-01-24 	2003-02-05 &	\cite{Kasamatsu2004}, \cite{Kasamatsu2005}, \cite{Preunkert2007} \\
206 & Kasamatsu &	Southern Ocean &	28 &	2002-02-13 	2002-02-28 &	\cite{Kasamatsu2004}, \cite{Kasamatsu2005}, \cite{Preunkert2007} \\
208 & Kasamatsu &	Southern Ocean (Indian sector) &	23 &	2008-02-05 	2008-02-22 &	unpublished \\
209 & Kiene &	NW Atlantic &	14 &	2002-04-06 	2002-04-19 	& \cite{Harada2004} \\
210 & Simo & E Mediterranean, Black Sea &	13 &	2000-10-06 	2000-10-28 &	\cite{Besiktepe2004} \\
211 & Matrai &	Arctic Ocean &	55 &	2001-07-11 	2001-08-17 &	\cite{Matrai2007}, \cite{Matrai2008} \\
212 & Matrai, Vernet & 	Balsfjord, Norway &	10 &	1999-03-26 	1999-05-18 &	unpublished \\
213 & Matrai &	 Sargasso Sea &	42 &	2004-07-15 	2004-08-01 &	\cite{Bailey2008}, \cite{Gabric2008} \\
214 & Belviso &  Northeast Atlantic Ocean &	78 &	2001-02-03 	2001-02-23 &	\cite{Belviso2004a}, \cite{Belviso2004b} \\
215 & Belviso &	 Northeast Atlantic Ocean &	80 &	2001-03-24 	2001-04-12 &	\cite{Belviso2004a}, \cite{Belviso2004b} \\
216 & Belviso &	 Northeast Atlantic Ocean &	74 &	2001-08-26 	2001-09-13 &	\cite{Belviso2004a}, \cite{Belviso2004b} \\
217 & Belviso &	Indian Sector of the Austral Ocean &	26 &	2005-01-18 	2005-02-12 &	\cite{Bopp2008}, \cite{Belviso2008} \\
218 & Shenoy &	Northern Arabian Sea &	15 &	1998-01-04 	1998-01-20 &	\cite{Shenoy2007} \\
219 & Shenoy &	Central Indian Ocean &	22 &	1998-02-22 	1998-03-28 &	\cite{Shenoy2002} \\
220 & Shenoy &	Arabian Sea (coastal) &	33 &	1998-07-20 	1998-08-14 &	\cite{Shenoy2007} \\
221 & Shenoy &	Bay of Bengal &	15 &	1998-10-30 	1998-11-05 &	\cite{Shenoy2000} \\
222 & Shenoy &	Arabian Sea &	19 &	1998-12-01 	1998-12-10 &	\cite{Shenoy2007} \\
223 & Shenoy &	Central Indian Ocean &	51 &	1999-01-27 	1999-03-10 &	\cite{Shenoy2002} \\
224 & Shenoy &	Bay of Bengal &	32 &	1999-07-27 	1999-08-05 &	\cite{Shenoy2007} \\
225 & Shenoy &	Bay of Bengal &	55 &	1999-08-11 	1999-08-31 &	\cite{Shenoy2007} \\
226 & Shenoy &	Arabian Sea (coastal) &	21 &	1999-09-10 	1999-10-10 &	\cite{Shenoy2007} \\
227 & Shenoy &	Southern Bay of Bengal &	42 &	2000-01-29 	2000-02-19 &	\cite{Shenoy2007} \\
228 & Shenoy &	Arabian Sea &	17 &	2000-11-06 	2000-11-14 &	\cite{Shenoy2007} \\
229 & Shenoy &	Bay of Bengal &	18 &	2001-07-10 	2001-08-01 &	\cite{Shenoy2006} \\
230 & Dacey &  Bermuda Atlantic &	126 &	1992-01-29 	1992-11-24 &	\cite{Dacey1998} \\
231 & Dacey &  Bermuda Atlantic &	276 &	1993-01-15 	1993-11-11 &	\cite{Dacey1998} \\
232 & Dacey & Bermuda Atlantic &	71 	&1994-01-18 	1994-11-15 &	\cite{Dacey1998} \\
233 & Saltzman & North Atlantic Ocean 	& 215 &	2007-07-17 	2007-07-24 &	\cite{Marandino2008} \\
234 & Lee, DiTullio, Balch & South Atlantic Ocean &	125 &	2008-12-05 	2008-12-31 &	unpublished \\
235 & Bange & Mauritanian coast &	43 &	2008-02-05 	2008-02-19 &	unpublished \\
236 & Simo 	& Antarctica &	53 &	2009-01-28 	2009-02-25 &	unpublished \\
237 & Borges, Sciare &	European Estuaries &	16 &	1998-05-25 	1998-05-29 &	\cite{Sciare2002} \\
238 & Borges, Sciare &	European Estuaries &	9 &	1996-12-09 	1996-12-11 &	\cite{Sciare2002} \\
239 & Borges, Sciare &	European Estuaries &	14 &	1996-07-08 	1996-07-11 &	\cite{Sciare2002} \\
240 & Borges, Sciare &	European Estuaries &	7 &	1997-06-13 	1997-06-15 &	\cite{Sciare2002} \\
241 & Borges, Sciare &	European Estuaries &	13 &	1997-09-01 	1997-09-04 &	\cite{Sciare2002} \\
242 & Matrai, Dacey, DiTullio, Kieber, Kiene, Simo & Western Antarctic Peninsula &	188 &	2005-11-17 	2006-02-27 &	unpublished \\
243 & Johnson &	Eastern Eq. Pacific &	1661 &	2008-10-20 	2008-11-22 &	unpublished  \\
\hline
\end{longtable}
\end{landscape}
}

\clearpage
\newpage

\begin{thebibliography}{136}
\providecommand{\natexlab}[1]{#1}
\expandafter\ifx\csname urlstyle\endcsname\relax
  \providecommand{\doi}[1]{doi:\discretionary{}{}{}#1}\else
  \providecommand{\doi}{doi:\discretionary{}{}{}\begingroup
  \urlstyle{rm}\Url}\fi

\bibitem[{\textit{Amouroux et~al.}(2002)\textit{Amouroux, Roberts,
  Rapsomanikis, and Andreae}}]{Amouroux2002}
Amouroux, D., G.~Roberts, S.~Rapsomanikis, and M.~O. Andreae (2002), Biogenic
  gas (CH4, N2O, DMS) emission to the atmosphere from near-shore and shelf
  waters of the North-Western Black Sea, \textit{Estuarine Coastal And Shelf
  Science}, \textit{54}(3), 575--587.

\bibitem[{\textit{Andreae}(1985)}]{Andreae1985}
Andreae, M.~O. (1985), Dimethylsulfide in the water column and the sediment
  porewaters of the Peru upwelling area, \textit{Limnology And Oceanography},
  \textit{30}(6), 1208--1218.

\bibitem[{\textit{Andreae and Barnard}(1984)}]{Andreae1984}
Andreae, M.~O., and W.~R. Barnard (1984), The marine chemistry of
  dimethylsulfide, \textit{Marine Chemistry}, \textit{14}(3), 267--279.

\bibitem[{\textit{Andreae and Raemdonck}(1983)}]{Andreae1983}
Andreae, M.~O., and H.~Raemdonck (1983), Dimethyl sulfide in the surface ocean
  and the marine atmosphere - a global view, \textit{Science},
  \textit{221}(4612), 744--747.

\bibitem[{\textit{Andreae et~al.}(1994)\textit{Andreae, Andreae, and
  Schebeske}}]{Andreae1994}
Andreae, T.~W., M.~O. Andreae, and G.~Schebeske (1994), Biogenic sulfur
  emissions and aerosols over the tropical South-Atlantic .1. dimethylsulfide
  in seawater and in the atmospheric boundary-layer, \textit{J. Geophys. Res.
  Atmos.}, \textit{99}(D11), 22,819--22,829.

\bibitem[{\textit{Bailey et~al.}(2008)}]{Bailey2008}
Bailey, K.~E., et~al. (2008), Dimethylsulfide production in Sargasso Sea
  eddies, \textit{Deep Sea Res. Part II}, \textit{55}(10-13), 1491--1504,
  \doi{10.1016/j.dsr2.2008.02.011}.

\bibitem[{\textit{Baker et~al.}(2000)\textit{Baker, Turner, Broadgate,
  Thompson, McFiggans, Vesperini, Nightingale, Liss, and Jickells}}]{Baker2000}
Baker, A.~R., S.~M. Turner, W.~J. Broadgate, A.~Thompson, G.~B. McFiggans,
  O.~Vesperini, P.~D. Nightingale, P.~S. Liss, and T.~D. Jickells (2000),
  Distribution and sea-air fluxes of biogenic trace gases in the Eastern
  Atlantic Ocean, \textit{Glob. Biogeochem. Cycles}, \textit{14}(3), 871--886.

\bibitem[{\textit{Barnard et~al.}(1982)\textit{Barnard, Andreae, Watkins,
  Bingemer, and Georgii}}]{Barnard1982}
Barnard, W.~R., M.~O. Andreae, W.~E. Watkins, H.~Bingemer, and H.~W. Georgii
  (1982), The flux of dimethylsulfide from the oceans to the atmosphere,
  \textit{Journal Of Geophysical Research Oceans And Atmospheres},
  \textit{87}(NC11), 8787--8793.

\bibitem[{\textit{Barnard et~al.}(1984)\textit{Barnard, Andreae, and
  IVERSON}}]{Barnard1984}
Barnard, W.~R., M.~O. Andreae, and R.~L. IVERSON (1984), Dimethylsulfide and
  Phaeocystis-poucheti in the Southeastern Bering Sea, \textit{Cont. Shelf
  Res.}, \textit{3}(2), 103--113.

\bibitem[{\textit{Bates and Cline}(1985)}]{Bates1985}
Bates, T.~S., and J.~D. Cline (1985), The role of the ocean in a regional
  sulfur cycle, \textit{J. Geophys. Res. Oceans}, \textit{90}(NC5), 9168--9172.

\bibitem[{\textit{Bates and Quinn}(1997)}]{Bates1997}
Bates, T.~S., and P.~K. Quinn (1997), Dimethylsulfide (DMS) in the equatorial
  Pacific Ocean (1982 to 1996): Evidence of a climate feedback?,
  \textit{Geophys. Res. Lett.}, \textit{24}, 861--864.

\bibitem[{\textit{Bates et~al.}(1987)\textit{Bates, Cline, Gammon, and
  Kellyhansen}}]{Bates1987}
Bates, T.~S., J.~D. Cline, R.~H. Gammon, and S.~R. Kellyhansen (1987), Regional
  and seasonal-variations in the flux of oceanic dimethylsulfide to the
  atmosphere, \textit{J. Geophys. Res. Oceans}, \textit{92}(C3), 2930--2938.

\bibitem[{\textit{Bates et~al.}(1990)\textit{Bates, Johnson, Quinn, Goldan,
  Kuster, Covert, and Hahn}}]{Bates1990}
Bates, T.~S., J.~E. Johnson, P.~K. Quinn, P.~D. Goldan, W.~C. Kuster, D.~C.
  Covert, and C.~J. Hahn (1990), The biogeochemical sulfur cycle in the marine
  boundary-layer over the Northeast Pacific-Ocean, \textit{Journal Of
  Atmospheric Chemistry}, \textit{10}(1), 59--81.

\bibitem[{\textit{Bates et~al.}(1992)\textit{Bates, Calhoun, and
  Quinn}}]{Bates1992}
Bates, T.~S., J.~A. Calhoun, and P.~K. Quinn (1992), Variations in the
  methanesulfonate to sulfate molar ratio in submicrometer marine
  aerosol-particles over the South-Pacific Ocean, \textit{J. Geophys. Res.
  Atmos.}, \textit{97}(D9), 9859--9865.

\bibitem[{\textit{Bates et~al.}(1993)\textit{Bates, Kelly, and
  Johnson}}]{Bates1993}
Bates, T.~S., K.~C. Kelly, and J.~E. Johnson (1993), Concentrations and fluxes
  of dissolved biogenic gases (DMS, CH4, CO, CO2) in the equatorial Pacific
  during the SAGA-3 experiment, \textit{J Geophys Res Atmos}, \textit{98}(D9),
  16,969--16,977.

\bibitem[{\textit{Bates et~al.}(1994)\textit{Bates, Kiene, Wolfe, Matrai,
  Chavez, Buck, Blomquist, and Cuhel}}]{Bates1994}
Bates, T.~S., R.~P. Kiene, G.~V. Wolfe, P.~A. Matrai, F.~P. Chavez, K.~R. Buck,
  B.~W. Blomquist, and R.~L. Cuhel (1994), The cycling of sulfur in surface
  seawater of the Northeast Pacific, \textit{J. Geophys. Res. Oceans},
  \textit{99}(C4), 7835--7843.

\bibitem[{\textit{Bates et~al.}(1998)\textit{Bates, Kapustin, Quinn, Covert,
  Coffman, Mari, Durkee, De~Bruyn, and Saltzman}}]{Bates1998}
Bates, T.~S., V.~N. Kapustin, P.~K. Quinn, D.~S. Covert, D.~J. Coffman,
  C.~Mari, P.~A. Durkee, W.~J. De~Bruyn, and E.~S. Saltzman (1998), Processes
  controlling the distribution of aerosol particles in the lower marine
  boundary layer during the first aerosol characterization experiment (ACE 1),
  \textit{J. Geophys. Res. Atmos.}, \textit{103}(D13), 16,369--16,383.

\bibitem[{\textit{Bates et~al.}(2001)}]{Bates2001}
Bates, T.~S., et~al. (2001), Regional physical and chemical properties of the
  marine boundary layer aerosol across the atlantic during AEROSOLS99: An
  overview, \textit{J. Geophys. Res. Atmos.}, \textit{106}(D18),
  20,767--20,782.

\bibitem[{\textit{Bell et~al.}(2006)\textit{Bell, Malin, McKee, and
  Liss}}]{Bell2006}
Bell, T.~G., G.~Malin, C.~M. McKee, and P.~S. Liss (2006), A comparison of
  dimethylsulphide (DMS) data from the Atlantic Meridional Transect (AMT)
  programme with proposed algorithms for global surface dms concentrations,
  \textit{Deep Sea Res. Part II}, \textit{53}(14-16), 1720--1735,
  \doi{10.1016/j.dsr2.2006.05.013}.

\bibitem[{\textit{Belviso et~al.}(1993)\textit{Belviso, Buatmenard, Putaud,
  Nguyen, Claustre, and Neveux}}]{Belviso1993}
Belviso, S., P.~Buatmenard, J.~P. Putaud, B.~C. Nguyen, H.~Claustre, and
  J.~Neveux (1993), Size distribution of dimethylsulfoniopropionate (DMSP) in
  areas of the Tropical Northeastern Atlantic-Ocean and the Mediterranean-Sea,
  \textit{Marine Chemistry}, \textit{44}(1), 55--71.

\bibitem[{\textit{Belviso et~al.}(2000)\textit{Belviso, Morrow, and
  Mihalopoulos}}]{Belviso2000}
Belviso, S., R.~Morrow, and N.~Mihalopoulos (2000), An Atlantic Meridional
  Transect of surface water dimethyl sulfide concentrations with 10-15 km
  horizontal resolution and close examination of ocean circulation, \textit{J.
  Geophys. Res. Atmos.}, \textit{105}(D11), 14,423--14,431.

\bibitem[{\textit{Belviso et~al.}(2003)\textit{Belviso, Sciandra, and
  Copin-Montegut}}]{Belviso2003}
Belviso, S., A.~Sciandra, and C.~Copin-Montegut (2003), Mesoscale features of
  surface water DMSP and DMS concentrations in the Atlantic Ocean off Morocco
  and in the Mediterranean Sea, \textit{Deep Sea Res. Part I}, \textit{50}(4),
  543--555.

\bibitem[{\textit{Belviso et~al.}(2004{\natexlab{a}})\textit{Belviso, Moulin,
  Bopp, and Stefels}}]{Belviso2004a}
Belviso, S., C.~Moulin, L.~Bopp, and J.~Stefels (2004{\natexlab{a}}),
  Assessment of a global climatology of oceanic dimethylsulfide (DMS)
  concentrations based on SeaWiFS imagery (1998-2001), \textit{Canadian Journal
  Of Fisheries And Aquatic Sciences}, \textit{61}(5), 804--816.

\bibitem[{\textit{Belviso et~al.}(2004{\natexlab{b}})}]{Belviso2004b}
Belviso, S., et~al. (2004{\natexlab{b}}), Comparison of global climatological
  maps of sea surface dimethyl sulfide, \textit{Global Biogeochem. Cycles},
  \textit{18}, GB3013, \doi{10.1029/2003GB002193}.

\bibitem[{\textit{Belviso et~al.}(2008)}]{Belviso2008}
Belviso, S., et~al. (2008), Effect of natural iron fertilisation on the
  distribution of DMS and DMSP in the Indian sector of the Southern Ocean,
  \textit{Deep Sea Res. Part II}, \textit{55}(5-7), 893--900.

\bibitem[{\textit{Berresheim}(1987)}]{Berresheim1987}
Berresheim, H. (1987), Biogenic sulfur emissions from the Sub-Antarctic and
  Antarctic Oceans, \textit{J. Geophys. Res. Atmos.}, \textit{92}(D11),
  13,245--13,262.

\bibitem[{\textit{Berresheim et~al.}(1989)\textit{Berresheim, Andreae, Ayers,
  and Gillett}}]{Berresheim1989}
Berresheim, H., M.~O. Andreae, G.~P. Ayers, and R.~W. Gillett (1989),
  Distribution of biogenic sulfur-compounds in the remote Southern-Hemisphere,
  \textit{Acs Symposium Series}, \textit{393}, 352--366.

\bibitem[{\textit{Berresheim et~al.}(1991)\textit{Berresheim, Andreae, Iverson,
  and Li}}]{Berresheim1991}
Berresheim, H., M.~O. Andreae, R.~L. Iverson, and S.~M. Li (1991),
  Seasonal-variations of dimethylsulfide emissions and atmospheric sulfur and
  nitrogen species over the Western North-Atlantic Ocean, \textit{Tellus Series
  B-Chemical And Physical Meteorology}, \textit{43}(5), 353--372.

\bibitem[{\textit{Berresheim et~al.}(1998)\textit{Berresheim, Huey, Thorn,
  Eisele, Tanner, and Jefferson}}]{Berresheim1998}
Berresheim, H., J.~W. Huey, R.~P. Thorn, F.~L. Eisele, D.~J. Tanner, and
  A.~Jefferson (1998), Measurements of dimethyl sulfide, dimethyl sulfoxide,
  dimethyl sulfone, and aerosol ions at Palmer station, Antarctica,
  \textit{Journal Of Geophysical Research-Atmospheres}, \textit{103}(D1),
  1629--1637.

\bibitem[{\textit{Besiktepe et~al.}(2004)\textit{Besiktepe, Tang, Vila, and
  Simo}}]{Besiktepe2004}
Besiktepe, S., K.~W. Tang, M.~Vila, and R.~Simo (2004), Dimethylated sulfur
  compounds in seawater, seston and mesozooplankton in the seas around turkey,
  \textit{Deep Sea Res. Part I}, \textit{51}(9), 1179--1197.

\bibitem[{\textit{Bingemer}(1984)}]{Bingemer1984}
Bingemer, H. (1984), Dimethylsulfid in ozean und mariner atmosphere -
  experimentelle untersuchung einer naturlichen schwefelquelle for die
  atmosphere., Ph.D. thesis, Univ. Frankfurt/Main, Frankfurt, Germany.

\bibitem[{\textit{Bingemer and Ockelmann}(1987)}]{Bingemer1987}
Bingemer, H. W. G. S.~B., H., and G.~Ockelmann (1987), \textit{Die Bedeutung
  biogener Schwefelemissionen for den tropospherischen Schwefelkreislauf, in
  Atmospherische Spurenstoffe}, Dtsch. Forsch., Weinheim. Germany.

\bibitem[{\textit{Blomquist et~al.}(1996)\textit{Blomquist, Bandy, and
  Thornton}}]{Blomquist1996}
Blomquist, B.~W., A.~R. Bandy, and D.~C. Thornton (1996), Sulfur gas
  measurements in the eastern north atlantic ocean during the atlantic
  stratocumulus transition experiment marine aerosol and gas exchange,
  \textit{J. Geophys. Res. Atmos.}, \textit{101}(D2), 4377--4392.

\bibitem[{\textit{Boniforti et~al.}(1993)\textit{Boniforti, Emaldi, Ferraroli,
  Maspero, Nair, and Novo}}]{Boniforti1993}
Boniforti, R., P.~Emaldi, R.~Ferraroli, M.~Maspero, R.~Nair, and A.~Novo
  (1993), Preliminary data on dms concentration in seawater samples collected
  from the la-spezia gulf (ligurian sea), \textit{Dimethylsulfide: Oceans,
  Atmosphere And Climate}, pp. 163--172.

\bibitem[{\textit{Bopp et~al.}(2008)\textit{Bopp, Aumont, Belviso, and
  Blain}}]{Bopp2008}
Bopp, L., O.~Aumont, S.~Belviso, and S.~Blain (2008), Modelling the effect of
  iron fertilization on dimethylsulphide emissions in the Southern Ocean,
  \textit{Deep Sea Research Part II-Topical Studies in Oceanography},
  \textit{55}(5-7), 901--912.

\bibitem[{\textit{Bouillon et~al.}(2002)\textit{Bouillon, Lee, de~Mora,
  Levasseur, and Lovejoy}}]{Bouillon2002}
Bouillon, R.~C., P.~A. Lee, S.~J. de~Mora, M.~Levasseur, and C.~Lovejoy (2002),
  Vernal distribution of dimethylsulphide, dimethylsulphoniopropionate, and
  dimethylsulphoxide in the North water in 1998, \textit{Deep-Sea Research Part
  II-Topical Studies in Oceanography}, \textit{49}(22-23), 5171--5189.

\bibitem[{\textit{Burgermeister et~al.}(1990)\textit{Burgermeister, Zimmermann,
  Georgii, Bingemer, Kirst, Janssen, and Ernst}}]{Burgermeister1990}
Burgermeister, S., R.~L. Zimmermann, H.~W. Georgii, H.~G. Bingemer, G.~O.
  Kirst, M.~Janssen, and W.~Ernst (1990), On the biogenic origin of
  dimethylsulfide - relation between Chlorophyll, ATP, organismic DMSP,
  phytoplankton species, and DMS distribution in Atlantic surface-water and
  atmosphere, \textit{Journal Of Geophysical Research-Atmospheres},
  \textit{95}(D12), 20,607--20,615.

\bibitem[{\textit{Cantin et~al.}(1996)\textit{Cantin, Levasseur, Gosselin, and
  Michaud}}]{Cantin1996}
Cantin, G., M.~Levasseur, M.~Gosselin, and S.~Michaud (1996), Role of
  zooplankton in the mesoscale distribution of surface dimethylsulfide
  concentrations in the Gulf of St Lawrence, Canada, \textit{Marine
  Ecology-Progress Series}, \textit{141}(1-3), 103--117.

\bibitem[{\textit{Chuck et~al.}(2005)\textit{Chuck, Turner, and
  Liss}}]{Chuck2005}
Chuck, A.~L., S.~M. Turner, and P.~S. Liss (2005), Oceanic distributions and
  air-sea fluxes of biogenic halocarbons in the open ocean, \textit{Journal Of
  Geophysical Research-Oceans}, \textit{110}(C10).

\bibitem[{\textit{Cline and Bates}(1983)}]{Cline1983}
Cline, J.~D., and T.~S. Bates (1983), Dimethyl sulfide in the equatorial
  Pacific-Ocean - a natural source of sulfur to the atmosphere,
  \textit{Geophysical Research Letters}, \textit{10}(10), 949--952.

\bibitem[{\textit{Curran et~al.}(1998)\textit{Curran, Jones, and
  Burton}}]{Curran1998}
Curran, M. A.~J., G.~B. Jones, and H.~Burton (1998), Spatial distribution of
  dimethylsulfide and dimethylsulfoniopropionate in the Australasian sector of
  the Southern Ocean, \textit{J. Geophys. Res.}, \textit{103}, 16,677--16,689.

\bibitem[{\textit{Dacey et~al.}(1998)\textit{Dacey, Howse, Michaels, and
  Wakeham}}]{Dacey1998}
Dacey, J. W.~H., F.~A. Howse, A.~F. Michaels, and S.~G. Wakeham (1998),
  Temporal variability of dimethylsulfide and dimethylsulfoniopropionate in the
  Sargasso Sea, \textit{Deep-Sea Research Part I-Oceanographic Research
  Papers}, \textit{45}(12), 2085--+.

\bibitem[{\textit{Deal et~al.}(2005)\textit{Deal, Kieber, Toole, Stamnes,
  Jiang, and Uzuka}}]{Deal2005}
Deal, C.~J., D.~J. Kieber, D.~A. Toole, K.~Stamnes, S.~Jiang, and N.~Uzuka
  (2005), Dimethylsulfide photolysis rates and apparent quantum yields in
  Bering Sea seawater, \textit{Continental Shelf Research}, \textit{25}(15),
  1825--1835.

\bibitem[{\textit{del Valle et~al.}(2007)\textit{del Valle, Kieber, Bisgrove,
  and Kiene}}]{Valle2007}
del Valle, D.~A., D.~J. Kieber, J.~Bisgrove, and R.~P. Kiene (2007),
  Light-stimulated production of dissolved dmso by a particle-associated
  process in the Ross Sea, Antarctica, \textit{Limnology And Oceanography},
  \textit{52}, 2456--2466.

\bibitem[{\textit{del Valle et~al.}(2009{\natexlab{a}})\textit{del Valle,
  Kieber, Toole, Bisgrove, and Kiene}}]{Valle2009}
del Valle, D.~A., D.~J. Kieber, D.~A. Toole, J.~Bisgrove, and R.~P. Kiene
  (2009{\natexlab{a}}), Dissolved DMSO production via biological and
  photochemical oxidation of dissolved DMS in the Ross Sea, Antarctica,
  \textit{Deep-Sea Research Part I-Oceanographic Research Papers},
  \textit{56}(2), 166--177.

\bibitem[{\textit{del Valle et~al.}(2009{\natexlab{b}})\textit{del Valle,
  Kieber, Toole, Brinkley, and Kiene}}]{Valle2009a}
del Valle, D.~A., D.~J. Kieber, D.~A. Toole, J.~Brinkley, and R.~P. Kiene
  (2009{\natexlab{b}}), Biological consumption of dimethylsulfide (DMS) and its
  importance in DMS dynamics in the Ross Sea, Antarctica, \textit{Limnology And
  Oceanography}, \textit{54}(3), 785--798.

\bibitem[{\textit{DiTullio and Smith}(1993)}]{DiTullio1993}
DiTullio, G.~R., and W.~O. Smith (1993), Dimethyl sulfide concentrations near
  the Antarctic Peninsula: November, 1992, Antarct., \textit{J. U. S.}, pp.
  130--132.

\bibitem[{\textit{DiTullio and Smith}(1995)}]{DiTullio1995}
DiTullio, G.~R., and W.~O. Smith (1995), Relationship between dimethylsulfide
  and phytoplankton pigment concentrations in the Ross Sea, Antarctica,
  \textit{Deep-Sea Research Part I-Oceanographic Research Papers},
  \textit{42}(6), 873--892.

\bibitem[{\textit{Froelich et~al.}(1985)\textit{Froelich, Kaul, Byrd, Andreae,
  and Roe}}]{Froelich1985}
Froelich, P.~N., L.~W. Kaul, J.~T. Byrd, M.~O. Andreae, and K.~K. Roe (1985),
  Arsenic, barium, germanium, tin, dimethylsulfide and nutrient biogeochemistry
  in Charlotte Harbor, Florida, a phosphorus-enriched estuary,
  \textit{Estuarine Coastal And Shelf Science}, \textit{20}(3), 239--264.

\bibitem[{\textit{Gabric et~al.}(2008)}]{Gabric2008}
Gabric, A.~J., et~al. (2008), Factors determining the vertical profile of
  dimethylsulfide in the Sargasso Sea during summer, \textit{Deep-Sea Research
  Part Ii-Topical Studies In Oceanography}, \textit{55}(10-13), 1505--1518.
  
\bibitem[{\textit{Gali and Simo}(2010)}]{Gali2010}
Gali, M., and R. Simo (2010), Occurrence and cycling of dimethylated sulfur 
	compounds in the Arctic during summer receding of the ice edge,
  \textit{Mar. Chem.},\doi{10.1016/j.marchem.2010.07.003}.
  
\bibitem[{\textit{Gibson et~al.}(1988)\textit{Gibson, Garrick, Burton, and
  McTaggart}}]{Gibson1988}
Gibson, J. A.~E., R.~C. Garrick, H.~R. Burton, and A.~R. McTaggart (1988),
  Dimethylsulfide concentrations in the ocean close to the Antarctic continent,
  \textit{Geomicrobiology Journal}, \textit{6}(3-4), 179--184.

\bibitem[{\textit{Groene}(1995)}]{Groene1995}
Groene, T. (1995), Biogenic production and consumption of dimethylsulfide (DMS)
  and dimethylsulfoniopropionate (DMSP) in the marine epipelagic zone - a
  review, \textit{Journal Of Marine Systems}, \textit{6}(3), 191--209.

\bibitem[{\textit{Harada et~al.}(2004)\textit{Harada, Rouse, Sunda, and
  Kiene}}]{Harada2004}
Harada, H., M.~A. Rouse, W.~Sunda, and R.~P. Kiene (2004), Latitudinal and
  vertical distributions of particle-associated dimethylsulfoniopropionate
  (DMSP) lyase activity in the Western North Atlantic Ocean, \textit{Canadian
  Journal Of Fisheries And Aquatic Sciences}, \textit{61}(5), 700--711.

\bibitem[{\textit{Hatton et~al.}(1996)\textit{Hatton, Malin, Turner, and
  Liss}}]{Hatton1996}
Hatton, A.~D., G.~Malin, S.~M. Turner, and P.~S. Liss (1996), DMSO - a
  significant compound in the biogeochemical cycle of DMS, \textit{Biological
  And Environmental Chemistry Of Dmsp And Related Sulfonium Compounds}, pp.
  405--412.

\bibitem[{\textit{Hatton et~al.}(1998)\textit{Hatton, Turner, Malin, and
  Liss}}]{Hatton1998}
Hatton, A.~D., S.~M. Turner, G.~Malin, and P.~S. Liss (1998),
  Dimethylsulphoxide and other biogenic sulphur compounds in the Galapagos
  Plume, \textit{Deep-Sea Research Part Ii-Topical Studies In Oceanography},
  \textit{45}(6), 1043--1053.

\bibitem[{\textit{Hatton et~al.}(1999)\textit{Hatton, Malin, and
  Liss}}]{Hatton1999}
Hatton, A.~D., G.~Malin, and P.~S. Liss (1999), Distribution of biogenic
  sulphur compounds during and just after the Southwest Monsoon in the Arabian
  Sea, \textit{Deep-Sea Research Part Ii-Topical Studies In Oceanography},
  \textit{46}(3-4), 617--632.

\bibitem[{\textit{Holligan et~al.}(1987)\textit{Holligan, Turner, and
  Liss}}]{Holligan1987}
Holligan, P., S.~Turner, and P.~Liss (1987), Measurements of dimethyl sulphide
  in frontal regions, \textit{Continental Shelf Research}, \textit{7}(2),
  213--224.

\bibitem[{\textit{Holligan et~al.}(1993)}]{Holligan1993}
Holligan, P.~M., et~al. (1993), A biogeochemical study of the coccolithophore,
  emiliania-huxleyi, in the North-Atlantic, \textit{Global Biogeochemical
  Cycles}, \textit{7}(4), 879--900.

\bibitem[{\textit{Hu et~al.}(1997)\textit{Hu, Tang, Li, and Yu}}]{Hu1997}
Hu, M., X.~Tang, J.~Li, and Z.~Yu (1997), Flux of dimethylsulfide in the
  Jiaozhou Bay, China., \textit{J. of Environ. Sci. (China)}, \textit{9},
  80--85.

\bibitem[{\textit{Huebert et~al.}(2004)\textit{Huebert, Blomquist, Hare,
  Fairall, Johnson, and Bates}}]{Huebert2004}
Huebert, B.~J., B.~W. Blomquist, J.~E. Hare, C.~W. Fairall, J.~E. Johnson, and
  T.~S. Bates (2004), Measurement of the sea-air DMS flux and transfer velocity
  using eddy correlation, \textit{Geophysical Research Letters},
  \textit{31}(23), L23,113.

\bibitem[{\textit{Iverson et~al.}(1989)\textit{Iverson, Nearhoof, and
  Andreae}}]{Iverson1989}
Iverson, R.~L., F.~L. Nearhoof, and M.~O. Andreae (1989), Production of
  dimethylsulfonium propionate and dimethylsulfide by phytoplankton in
  estuarine and coastal waters, \textit{Limnology And Oceanography},
  \textit{34}(1), 53--67.

\bibitem[{\textit{Jickells et~al.}(2008)}]{Jickells2008}
Jickells, T.~D., et~al. (2008), A lagrangian biogeochemical study of an eddy in
  the Northeast Atlantic, \textit{Progress In Oceanography}, \textit{76}(3),
  366--398.

\bibitem[{\textit{Kasamatsu et~al.}(2004)\textit{Kasamatsu, Kawaguchi,
  Watanabe, Odate, and Fukuchi}}]{Kasamatsu2004}
Kasamatsu, N., S.~Kawaguchi, S.~Watanabe, T.~Odate, and M.~Fukuchi (2004),
  Possible impacts of zooplankton grazing on dimethylsulfide production in the
  Antarctic Ocean, \textit{Canadian Journal Of Fisheries And Aquatic Sciences},
  \textit{61}(5), 736--743.

\bibitem[{\textit{Kasamatsu et~al.}(2005)\textit{Kasamatsu, T., and
  Fukuchi}}]{Kasamatsu2005}
Kasamatsu, N., O.~T., and M.~Fukuchi (2005), Dimethylsulfide and
  dimethylsulfoniopropionate production in the Antarctic pelagic food web,
  \textit{Ocean and Polar Research}, \textit{27}, 197--203.

\bibitem[{\textit{Kieber et~al.}(1996)\textit{Kieber, Jiao, Kiene, and
  Bates}}]{Kieber1996}
Kieber, D.~J., J.~F. Jiao, R.~P. Kiene, and T.~S. Bates (1996), Impact of
  dimethylsulfide photochemistry on methyl sulfur cycling in the equatorial
  pacific ocean, \textit{Journal Of Geophysical Research-Oceans},
  \textit{101}(C2), 3715--3722.

\bibitem[{\textit{Kiene}(1996)}]{Kiene1996}
Kiene, R.~P. (1996), \textit{Turnover of dissolved DMSP in the estuarine and
  shelf waters of the northern Gulf of Mexico, in Biological and Environmental
  Chemistry of DMSP and Related Sulfonium Compounds}, Plenum, New York.

\bibitem[{\textit{Kiene et~al.}(2007)\textit{Kiene, Kieber, Slezak, Toole, del
  Valle, Bisgrove, Brinkley, and Rellinger}}]{Kiene2007}
Kiene, R.~P., D.~J. Kieber, D.~Slezak, D.~A. Toole, D.~A. del Valle,
  J.~Bisgrove, J.~Brinkley, and A.~Rellinger (2007), Distribution and cycling
  of dimethylsulfide, dimethylsulfoniopropionate, and dimethylsulfoxide during
  spring and early summer in the Southern Ocean South of New Zealand,
  \textit{Aquatic Sciences}, \textit{69}(3), 305--319.

\bibitem[{\textit{Kirst et~al.}(1993)\textit{Kirst, Wanzek, Haase,
  Rapsomanikis, de~Mora, Schebeske, and Andreae}}]{Kirst1993}
Kirst, G.~O., M.~Wanzek, R.~Haase, S.~Rapsomanikis, S.~de~Mora, G.~Schebeske,
  and M.~O. Andreae (1993), \textit{Ecophysiology of ice algae (Antarctica):
  Dimethylsulfoniopropionate content and release of dirnethylsulfide during ice
  melt, in Dimethylsulphide: Oceans, Atmosphere, and Climate.}, KIuwer,
  Dordrecht. Germany.

\bibitem[{\textit{Kwint and Kramer}(1996)}]{Kwint1996}
Kwint, R. L.~J., and K.~J.~M. Kramer (1996), Annual cycle of the production and
  fate of DMS and DMSP in a marine coastal system, \textit{Marine
  Ecology-Progress Series}, \textit{134}(1-3), 217--224.

\bibitem[{\textit{Leck and Persson}(1996)}]{Leck1996}
Leck, C., and C.~Persson (1996), The central Arctic ocean as a source of
  dimethyl sulfide - seasonal variability in relation to biological activity,
  \textit{Tellus Series B-Chemical And Physical Meteorology}, \textit{48}(2),
  156--177.

\bibitem[{\textit{Leck and Rodhe}(1991)}]{Leck1991}
Leck, C., and H.~Rodhe (1991), Emissions of marine biogenic sulfur to the
  atmosphere of Northern Europe, \textit{Journal of Atmospheric Chemistry},
  \textit{12}(1), 63--86.

\bibitem[{\textit{Leck et~al.}(1990)\textit{Leck, Larsson, Bågander,
  Johansson, and Hajdu}}]{Leck1990}
Leck, C., U.~Larsson, L.~E. Bågander, S.~Johansson, and S.~Hajdu (1990),
  Dimethyl sulfide in the Baltic Sea: Annual variability in relation to
  biological activity, \textit{J. Geophys. Res.}, \textit{95}, 3353--3363.

\bibitem[{\textit{Ledyard and Dacey}(1996)}]{Ledyard1996}
Ledyard, K.~M., and J.~W.~H. Dacey (1996), \textit{Kinetics of DMSP-Iyase
  activity in coastal seawater, in Biological and Environmental Chemistry of
  DMSP and Related Sulfonium Compounds}, Plenum, New York,.

\bibitem[{\textit{Lee and deMora}(1996)}]{Lee1996}
Lee, P.~A., and S.~J. deMora (1996), DMSP, DMS and DMSO concentrations and
  temporal trends in marine surface waters at leigh, New Zealand,
  \textit{Biological And Environmental Chemistry Of Dmsp And Related Sulfonium
  Compounds}, pp. 391--404.

\bibitem[{\textit{Liss et~al.}(1993)\textit{Liss, Watson, Liddicoat, Malin,
  Nightingale, Turner, and Upstillgoddard}}]{Liss1993}
Liss, P.~S., A.~J. Watson, M.~I. Liddicoat, G.~Malin, P.~D. Nightingale, S.~M.
  Turner, and R.~C. Upstillgoddard (1993), Trace gases and air-sea exchanges,
  \textit{Philosophical Transactions Of The Royal Society Of London Series
  A-Mathematical Physical And Engineering Sciences}, \textit{343}(1669),
  531--541.

\bibitem[{\textit{Liss et~al.}(1997)\textit{Liss, Hatton, Malin, Nightingale,
  and Turner}}]{Liss1997}
Liss, P.~S., A.~D. Hatton, G.~Malin, P.~D. Nightingale, and S.~M. Turner
  (1997), Marine sulphur emissions, \textit{Philosophical Transactions Of The
  Royal Society B-Biological Sciences}, \textit{352}(1350), 159--168.

\bibitem[{\textit{Lizotte et~al.}(2008)\textit{Lizotte, Levasseur, Scarratt,
  Michaud, Merzouk, Gosselin, and Pommier}}]{Lizotte2008}
Lizotte, M., M.~Levasseur, M.~G. Scarratt, S.~Michaud, A.~Merzouk, M.~Gosselin,
  and J.~Pommier (2008), Fate of dimethylsulfoniopropionate (DMSP) during the
  decline of the Northwest Atlantic Ocean spring diatom bloom, \textit{Aquatic
  Microbial Ecology}, \textit{52}(2), 159--173.

\bibitem[{\textit{Locarnini et~al.}(1998)\textit{Locarnini, Turner, and
  Liss}}]{Locarnini1998}
Locarnini, S. J.~P., S.~M. Turner, and P.~S. Liss (1998), The distribution of
  dimethylsulfide, DMS, and dimethylsulfoniopropionate, DMSP, in waters off the
  Western coast of Ireland, \textit{Continental Shelf Research},
  \textit{18}(12), 1455--1473.

\bibitem[{\textit{Lovelock et~al.}(1972)\textit{Lovelock, Maggs, and
  Rasmusse.RA}}]{Lovelock1972}
Lovelock, J.~E., R.~J. Maggs, and Rasmusse.RA (1972), Atmospheric dimethyl
  sulfide and natural sulfur cycle, \textit{Nature}, \textit{237}(5356),
  452--\&.

\bibitem[{\textit{Marandino et~al.}(2007)\textit{Marandino, De~Bruyn, Miller,
  and Saltzman}}]{Marandino2007}
Marandino, C.~A., W.~J. De~Bruyn, S.~D. Miller, and E.~S. Saltzman (2007), Eddy
  correlation measurements of the air/sea flux of dimethylsulfide over the
  North Pacific Ocean, \textit{Journal Of Geophysical Research-Atmospheres},
  \textit{112}(D3), D03,301.

\bibitem[{\textit{Marandino et~al.}(2008)\textit{Marandino, De~Bruyn, Miller,
  and Saltzman}}]{Marandino2008}
Marandino, C.~A., W.~J. De~Bruyn, S.~D. Miller, and E.~S. Saltzman (2008), DMS
  air/sea flux and gas transfer coefficients from the North Atlantic summertime
  coccolithophore bloom, \textit{Geophysical Research Letters},
  \textit{35}(23), L23,812.

\bibitem[{\textit{Marandino et~al.}(2009)\textit{Marandino, De~Bruyn, Miller,
  and Saltzman}}]{Marandino2009}
Marandino, C.~A., W.~J. De~Bruyn, S.~D. Miller, and E.~S. Saltzman (2009), Open
  ocean DMS air/sea fluxes over the Eastern South Pacific Ocean, \textit{Atmos.
  Chem. Phys.}, \textit{9}(2), 345--356.

\bibitem[{\textit{Matrai et~al.}(2007)\textit{Matrai, Vernet, and
  Wassmann}}]{Matrai2007}
Matrai, P., M.~Vernet, and P.~Wassmann (2007), Relating temporal and spatial
  patterns of DMSP in the Barents Sea to phytoplankton biomass and
  productivity, \textit{J. Mar. Syst.}, \textit{67}(1-2), 83--101,
  \doi{10.1016/j.jmarsys.2006.10.001}.

\bibitem[{\textit{Matrai and Keller}(1993)}]{Matrai1993}
Matrai, P.~A., and M.~D. Keller (1993), Dimethylsulfide in a large-scale
  coccolithophore bloom in the Gulf of Maine, \textit{Continental Shelf
  Research}, \textit{13}(8-9), 831--843.

\bibitem[{\textit{Matrai and Vernet}(1997)}]{Matrai1997}
Matrai, P.~A., and M.~Vernet (1997), Dynamics of the vernal bloom in the
  marginal ice zone of the Barents Sea: Dimethyl sulfide and
  dimethylsulfoniopropionate budgets, \textit{Journal Of Geophysical
  Research-Oceans}, \textit{102}(C10), 22,965--22,979.

\bibitem[{\textit{Matrai et~al.}(1993)\textit{Matrai, Balch, Cooper, and
  Saltzman}}]{Matrai1993a}
Matrai, P.~A., W.~M. Balch, D.~J. Cooper, and E.~S. Saltzman (1993), Ocean
  color and atmospheric dimethyl sulfide - on their mesoscale variability,
  \textit{Journal Of Geophysical Research-Atmospheres}, \textit{98}(D12),
  23,469--23,476.

\bibitem[{\textit{Matrai et~al.}(1996)\textit{Matrai, Cooper, and
  Saltzman}}]{Matrai1996}
Matrai, P.~A., D.~J. Cooper, and E.~S. Saltzman (1996), Frontal enhancement of
  dimethylsulfide concentrations across a gulf stream meander, \textit{Journal
  Of Marine Systems}, \textit{7}(1), 1--8.

\bibitem[{\textit{Matrai et~al.}(2008)\textit{Matrai, Tranvik, Leck, and
  Knulst}}]{Matrai2008}
Matrai, P.~A., L.~Tranvik, C.~Leck, and J.~C. Knulst (2008), Are high Arctic
  surface microlayers a potential source of aerosol organic precursors?,
  \textit{Marine Chemistry}, \textit{108}(1-2), 109--122.

\bibitem[{\textit{McTaggart and Burton}(1992)}]{McTaggart1992}
McTaggart, A.~R., and H.~Burton (1992), Dimethyl sulfide concentrations in the
  surface waters of the Australasian Antarctic and Sub-Antarctic oceans during
  an austral summer, \textit{Journal Of Geophysical Research-Oceans},
  \textit{97}(C9), 14,407--14,412.

\bibitem[{\textit{Merzouk et~al.}(2008)\textit{Merzouk, Levasseur, Scarratt,
  Michaud, Lizotte, Rivkin, and Kiene}}]{Merzouk2008}
Merzouk, A., M.~Levasseur, M.~Scarratt, S.~Michaud, M.~Lizotte, R.~B. Rivkin,
  and R.~P. Kiene (2008), Bacterial DMSP metabolism during the senescence of
  the spring diatom bloom in the Northwest Atlantic, \textit{Marine
  Ecology-Progress Series}, \textit{369}, 1--11, \doi{10.3354/meps07664}.

\bibitem[{\textit{Mihalopoulos}(1989)}]{Mihalopoulos1989}
Mihalopoulos, N. (1989), Contribution a l'etude du cycle biogeochimique de
  i'oxysulfure de carbone dans i'atmosphere, Ph.D. thesis, Univ. de Paris Vll,
  Paris.

\bibitem[{\textit{Mihalopoulos et~al.}(1992)\textit{Mihalopoulos, Nguyen,
  Putaud, and Belviso}}]{Mihalopoulos1992}
Mihalopoulos, N., B.~C. Nguyen, J.~P. Putaud, and S.~Belviso (1992), The
  oceanic source of carbonyl sulfide (COS), \textit{Atmospheric Environment
  Part A-General Topics}, \textit{26}(8), 1383--1394.

\bibitem[{\textit{Nguyen et~al.}(1978)\textit{Nguyen, Gaudry, Bonsang, and
  Lambert}}]{Nguyen1978}
Nguyen, B.~C., A.~Gaudry, B.~Bonsang, and G.~Lambert (1978), Re-evaluation of
  role of dimethyl sulfide in sulfur budget, \textit{Nature},
  \textit{275}(5681), 637--639.

\bibitem[{\textit{Nguyen et~al.}(1990)\textit{Nguyen, Mihalopoulos, and
  Belviso}}]{Nguyen1990}
Nguyen, B.~C., N.~Mihalopoulos, and S.~Belviso (1990), Seasonal-variation of
  atmospheric dimethylsulfide at Amsterdam Island in the Southern Indian-ocean,
  \textit{Journal Of Atmospheric Chemistry}, \textit{11}(1-2), 123--141.

\bibitem[{\textit{Nguyen et~al.}(1992)\textit{Nguyen, Mihalopoulos, Putaud,
  Gaudry, Gallet, Keene, and Galloway}}]{Nguyen1992}
Nguyen, B.~C., N.~Mihalopoulos, J.~P. Putaud, A.~Gaudry, L.~Gallet, W.~C.
  Keene, and J.~N. Galloway (1992), Covariations in oceanic dimethyl sulfide,
  its oxidation-products and rain acidity at Amsterdam Island in the Southern
  Indian-ocean, \textit{Journal Of Atmospheric Chemistry}, \textit{15}(1),
  39--53.

\bibitem[{\textit{Pfannkuche et~al.}(1993)\textit{Pfannkuche, Duinker, Graf,
  Henrich, Thiel, , and Zeitzschel}}]{Pfannkuche1993}
Pfannkuche, O., J.~C. Duinker, G.~Graf, R.~Henrich, H.~Thiel, , and
  B.~Zeitzschel (1993), Meteor-berichte, in \textit{Inst. for Meereskunde der
  Univ. Hamburg, Hamburg, Germany}.

\bibitem[{\textit{Preunkert et~al.}(2007)\textit{Preunkert, Legrand, Jourdain,
  Moulin, Belviso, Kasamatsu, Fukuchi, and Hirawake}}]{Preunkert2007}
Preunkert, S., M.~Legrand, B.~Jourdain, C.~Moulin, S.~Belviso, N.~Kasamatsu,
  M.~Fukuchi, and T.~Hirawake (2007), Interannual variability of
  dimethylsulfide in air and seawater and its atmospheric oxidation by-products
  (methanesulfonate and sulfate) at Dumont d'Urville, Coastal Antarctica
  (1999-2003), \textit{Journal Of Geophysical Research-Atmospheres},
  \textit{112}(D6), D06,306.

\bibitem[{\textit{Putaud and Nguyen}(1996)}]{Putaud1996}
Putaud, J.~P., and B.~C. Nguyen (1996), Assessment of dimethylsulfide sea-air
  exchange rate, \textit{Journal Of Geophysical Research-Atmospheres},
  \textit{101}(D2), 4403--4411.

\bibitem[{\textit{Putaud et~al.}(1993{\natexlab{a}})\textit{Putaud, Nguyen,
  Belviso, and Mihalopoulos}}]{Putaud1993b}
Putaud, J.~P., B.~Nguyen, S.~Belviso, and N.~Mihalopoulos (1993{\natexlab{a}}),
  \textit{Are dirnethylsulfide and condensation nuclei connected over the
  tropical Northeastern Atlantic Ocean? in Dimethylsulphide: Oceans,
  Atmosphere, and Climate.}, Kluwer, Dordrecht, Germany.

\bibitem[{\textit{Putaud et~al.}(1993{\natexlab{b}})\textit{Putaud, Belviso,
  Nguyen, and Mihalopoulos}}]{Putaud1993a}
Putaud, J.~P., S.~Belviso, B.~C. Nguyen, and N.~Mihalopoulos
  (1993{\natexlab{b}}), Dimethylsulfide, aerosols, and condensation nuclei over
  the tropical Northeastern Atlantic-ocean, \textit{Journal Of Geophysical
  Research-Atmospheres}, \textit{98}(D8), 14,863--14,871.

\bibitem[{\textit{Quinn et~al.}(1990)\textit{Quinn, Bates, Johnson, Covert, and
  Charlson}}]{Quinn1990}
Quinn, P.~K., T.~S. Bates, J.~E. Johnson, D.~S. Covert, and R.~J. Charlson
  (1990), Interactions between the sulfur and reduced nitrogen cycles over the
  central Pacific-ocean, \textit{Journal Of Geophysical Research-Atmospheres},
  \textit{95}(D10), 16,405--16,416.

\bibitem[{\textit{Scarratt et~al.}(2007)\textit{Scarratt, Levasseur, Michaud,
  and Roy}}]{Scarratt2007}
Scarratt, M., M.~Levasseur, S.~Michaud, and S.~Roy (2007), DMSP and DMS in the
  Northwest Atlantic: Late-summer distributions, production rates and sea-air
  fluxes, \textit{Aquatic Sciences - Research Across Boundaries},
  \textit{69}(3), 292--304.

\bibitem[{\textit{Sciare et~al.}(1999)\textit{Sciare, Mihalopoulos, and
  Nguyen}}]{Sciare1999}
Sciare, J., N.~Mihalopoulos, and B.~Nguyen (1999), Summertime seawater
  concentrations of dimethylsulfide in the Western Indian ocean: Reconciliation
  of fluxes and spatial variability with long-term atmospheric observations,
  \textit{Journal of Atmospheric Chemistry}, \textit{32}, 357--373(17).

\bibitem[{\textit{Sciare et~al.}(2002)\textit{Sciare, Mihalopoulos, and
  Nguyen}}]{Sciare2002}
Sciare, J., N.~Mihalopoulos, and B.~C. Nguyen (2002), Spatial and temporal
  variability of dissolved sulfur compounds in European estuaries,
  \textit{Biogeochemistry}, \textit{59}(1-2), 121--141.

\bibitem[{\textit{Sharma et~al.}(1999)\textit{Sharma, Barrie, Plummer,
  McConnell, Brickell, Levasseur, Gosselin, and Bates}}]{Sharma1999}
Sharma, S., L.~A. Barrie, D.~Plummer, J.~C. McConnell, P.~C. Brickell,
  M.~Levasseur, M.~Gosselin, and T.~S. Bates (1999), Flux estimation of oceanic
  dimethyl sulfide around North America, \textit{Journal Of Geophysical
  Research-Atmospheres}, \textit{104}(D17), 21,327--21,342.

\bibitem[{\textit{Shenoy and Kumar}(2007)}]{Shenoy2007}
Shenoy, D.~M., and M.~D. Kumar (2007), Variability in abundance and fluxes of
  dimethyl sulphide in the Indian Ocean, \textit{Biogeochemistry},
  \textit{83}(1-3), 277--292, \doi{10.1007/s10533-007-9092-4}.

\bibitem[{\textit{Shenoy et~al.}(2000)\textit{Shenoy, Kumar, and
  Sarma}}]{Shenoy2000}
Shenoy, D.~M., M.~D. Kumar, and V.~V. S.~S. Sarma (2000), Controls of dimethyl
  sulphide in the Bay of Bengal during bobmex-pilot cruise 1998,
  \textit{Proceedings Of The Indian Academy Of Sciences-Earth And Planetary
  Sciences}, \textit{109}(2), 279--283.

\bibitem[{\textit{Shenoy et~al.}(2002)\textit{Shenoy, Joseph, Kumar, and
  George}}]{Shenoy2002}
Shenoy, D.~M., S.~Joseph, M.~D. Kumar, and M.~D. George (2002), Control and
  interannual variability of dimethyl sulfide in the Indian Ocean,
  \textit{Journal Of Geophysical Research-Atmospheres}, \textit{107}(D19),
  8008.

\bibitem[{\textit{Shenoy et~al.}(2006)\textit{Shenoy, Paul, Gauns, Ramaiah, and
  Kumar}}]{Shenoy2006}
Shenoy, D.~M., J.~T. Paul, M.~Gauns, N.~Ramaiah, and M.~D. Kumar (2006),
  Spatial variations of DMS, DMSP and phytoplankton in the Bay of Bengal during
  the summer monsoon 2001, \textit{Marine Environmental Research},
  \textit{62}(2), 83--97.

\bibitem[{\textit{Siegel and Michaels}(1996)}]{Siegel1996}
Siegel, D.~A., and A.~F. Michaels (1996), Quantification of non-algal light
  attenuation in the Sargasso Sea: Implications for biogeochemistry and remote
  sensing, \textit{Deep-Sea Research Part Ii-Topical Studies In Oceanography},
  \textit{43}(2-3), 321--345.

\bibitem[{\textit{Simo et~al.}(1995)\textit{Simo, Grimalt, Pedros-Alio, and
  Albaiges}}]{Simo1995}
Simo, R., J.~O. Grimalt, C.~Pedros-Alio, and J.~Albaiges (1995), Occurrence and
  transformation of dissolved dimethyl sulfur species in stratified seawater
  (Western Mediterranean-Sea), \textit{Marine Ecology-Progress Series},
  \textit{127}(1-3), 291--299.

\bibitem[{\textit{Simo et~al.}(1997)\textit{Simo, Grimalt, and
  Albaiges}}]{Simo1997}
Simo, R., J.~O. Grimalt, and J.~Albaiges (1997), Dissolved dimethylsulphide,
  dimethylsulphoniopropionate and dimethylsulphoxide in Western Mediterranean
  waters, \textit{Deep-Sea Research Part Ii-Topical Studies In Oceanography},
  \textit{44}(3-4), 929--950.

\bibitem[{\textit{Smythe-Wright et~al.}(2005)\textit{Smythe-Wright, Boswell,
  Lucas, New, and Varney}}]{Smythe-Wright2005}
Smythe-Wright, D., S.~M. Boswell, C.~H. Lucas, A.~L. New, and M.~S. Varney
  (2005), Halocarbon and dimethyl sulphide studies around the mascarene
  plateau, \textit{Philosophical Transactions Of The Royal Society Of London
  Series A-Mathematical Physical And Engineering Sciences}, \textit{363}(1826),
  169--185.

\bibitem[{\textit{Staubes and Georgii}(1993)}]{Staubes1993a}
Staubes, R., and H.~W. Georgii (1993), Biogenic sulfur-compounds in seawater
  and the atmosphere of the Antarctic region, \textit{Tellus Series B-Chemical
  And Physical Meteorology}, \textit{45}(2), 127--137.

\bibitem[{\textit{Staubes-Diederich}(1992)}]{Staubes-Diederich1992}
Staubes-Diederich, R. (1992), Veneilung von dimethylsulfid, carbonylsulfid und
  schwefelkohlenstoff in ozean und mariner atmosphere, Ph.D. thesis, Johann
  Wolfgang Goethe-Universitat, Frankfurt am Main, Gernany.

\bibitem[{\textit{Stauhes and Georgii}(1993)}]{Staubes1993b}
Stauhes, R., and H.-W. Georgii (1993), \textit{Measurements of atmospheric and
  seawater DMS concentrations in the Atlantic, the Arctic, and the Antarctic
  region, in Dimethylsulfide: Oceans, Atmosphere, and Climate.}, G. Restelli
  and G. Angeletti.

\bibitem[{\textit{Tanzer and Heumann}(1992)}]{Tanzer1992}
Tanzer, D., and K.~G. Heumann (1992), Gas-chromatographic trace-level
  determination of volatile organic sulfides and selenides and of methyl-iodide
  in Atlantic surface-water, \textit{International Journal Of Environmental
  Analytical Chemistry}, \textit{48}(1), 17--31.

\bibitem[{\textit{Turner et~al.}(1988)\textit{Turner, Malin, Liss, Harbour, and
  Holligan}}]{Turner1988}
Turner, S.~M., G.~Malin, P.~S. Liss, D.~S. Harbour, and P.~M. Holligan (1988),
  The seasonal-variation of dimethyl sulfide and dimethylsulfoniopropionate
  concentrations in nearshore waters, \textit{Limnology And Oceanography},
  \textit{33}(3), 364--375.

\bibitem[{\textit{Turner et~al.}(1989)\textit{Turner, Malin, and
  Liss}}]{Turner1989}
Turner, S.~M., G.~Malin, and P.~S. Liss (1989), \textit{Dimethyl sulfide and
  (dimethylsulfonio) propionate in European coastal and shelf waters, in
  Biogenic Sulfur in the Environment}, Am. Chem. Soc., Washington, D. C.

\bibitem[{\textit{Turner et~al.}(1995)\textit{Turner, Nightingale, Broadgate,
  and Liss}}]{Turner1995}
Turner, S.~M., P.~D. Nightingale, W.~Broadgate, and P.~S. Liss (1995), The
  distribution of dimethyl sulfide and dimethylsulphoniopropionate in Antarctic
  waters and sea-ice, \textit{Deep-Sea Research Part Ii-Topical Studies In
  Oceanography}, \textit{42}(4-5), 1059--1080.

\bibitem[{\textit{Turner et~al.}(1996{\natexlab{a}})\textit{Turner, Malin,
  Nightingale, and Liss}}]{Turner1996a}
Turner, S.~M., G.~Malin, P.~D. Nightingale, and P.~S. Liss
  (1996{\natexlab{a}}), Seasonal variation of dimethyl sulphide in the North
  Sea and an assessment of fluxes to the atmosphere, \textit{Marine Chemistry},
  \textit{54}(3-4), 245--262.

\bibitem[{\textit{Turner et~al.}(1996{\natexlab{b}})\textit{Turner,
  Nightingale, Spokes, Liddicoat, and Liss}}]{Turner1996b}
Turner, S.~M., P.~D. Nightingale, L.~J. Spokes, M.~I. Liddicoat, and P.~S. Liss
  (1996{\natexlab{b}}), Increased dimethyl sulphide concentrations in sea water
  from in situ iron enrichment, \textit{Nature}, \textit{383}(6600), 513--517.

\bibitem[{\textit{Turner et~al.}(2004)\textit{Turner, Harvey, Law, Nightingale,
  and Liss}}]{Turner2004}
Turner, S.~M., M.~J. Harvey, C.~S. Law, P.~D. Nightingale, and P.~S. Liss
  (2004), Iron-induced changes in oceanic sulfur biogeochemistry,
  \textit{Geophysical Research Letters}, \textit{31}(14), L14,307.

\bibitem[{\textit{Uchida et~al.}(1992)\textit{Uchida, Ooguri, and
  Ishida}}]{Uchida1992}
Uchida, A., T.~Ooguri, and Y.~Ishida (1992), The distribution of
  dimethylsulfide in the waters off Japan and in the subtropical and tropical
  Pacific-Ocean, \textit{Nippon Suisan Gakkaishi}, \textit{58}(2), 261--265.

\bibitem[{\textit{Uher et~al.}(1995)\textit{Uher, Schebeske, Rapsomanikis, and
  Andreae}}]{Uher1995}
Uher, G., G.~Schebeske, S.~Rapsomanikis, and M.~O. Andreae (1995), Measurements
  of dimethyl sulfide in surface waters of the Northeast Atlantic ocean, in
  \textit{Ann. Geophys., 13. suppl.II, Part II, C394, Hamburg, Germany}.

\bibitem[{\textit{Uher et~al.}(1996)\textit{Uher, Schebeske, Rapsomanikis, and
  Andreae}}]{Uher1996}
Uher, G., G.~Schebeske, S.~Rapsomanikis, and M.~O. Andreae (1996), The
  distribution of dissolved dimethyl sulfide at the European continental
  margin, in \textit{Ann. Geophys., 14. suppl. II, C590, Den Haag,
  Netherlands}.

\bibitem[{\textit{Uher et~al.}(1997)\textit{Uher, Schebeske, Rapsomanikis, and
  Andreae}}]{Uher1997}
Uher, G., G.~Schebeske, S.~Rapsomanikis, and M.~O. Andreae (1997), The
  distribution of dissolved dimethylsulfide at the european continental
  margin., in \textit{OMEX I final report, Subproject Carbon Cycling and
  Biogases, Brussels, European Union Marine Science and Technology (MAST)
  program}.

\bibitem[{\textit{Uzuka et~al.}(1996)\textit{Uzuka, Watanabe, and
  Tsunogai}}]{Uzuka1996}
Uzuka, N., S.~Watanabe, and S.~Tsunogai (1996), Dimethylsulfide in coastal zone
  of the East China sea, \textit{Journal of Oceanography}, \textit{52}(3),
  313--321.

\bibitem[{\textit{Vallina and Simo}(2007)}]{Vallina2007}
Vallina, S.~M., and R.~Simo (2007), Strong relationship between DMS and the
  solar radiation dose over the global surface ocean, \textit{Science},
  \textit{315}(5811), 506--508.

\bibitem[{\textit{Vila-Costa et~al.}(2008)\textit{Vila-Costa, Kiene, and
  Simo}}]{Vila-Costa2008}
Vila-Costa, M., R.~P. Kiene, and R.~Simo (2008), Seasonal variability of the
  dynamics of dimethylated sulfur compounds in a coastal northwest
  mediterranean site, \textit{Limnology And Oceanography}, \textit{53}(1),
  198--211.

\bibitem[{\textit{Watanabe et~al.}(1995)\textit{Watanabe, Yamamoto, and
  Tsunogai}}]{Watanabe1995b}
Watanabe, S., H.~Yamamoto, and S.~Tsunogai (1995), Relation between the
  concentrations of DMS in surface seawater and air in the temperate North
  Pacific region, \textit{Journal Of Atmospheric Chemistry}, \textit{22}(3),
  271--283.

\bibitem[{\textit{Wong et~al.}(2005)\textit{Wong, Wong, Richardson, Smith,
  Arychuk, and Page}}]{Wong2005}
Wong, C.~S., S.~E. Wong, W.~A. Richardson, G.~E. Smith, M.~D. Arychuk, and
  J.~S. Page (2005), Temporal and spatial distribution of dimethylsulfide in
  the Subarctic Northeast Pacific ocean: a high-nutrient-low-chlorophyll
  region, \textit{Tellus B}, \textit{57}(4), 317--331.

\bibitem[{\textit{Wong et~al.}(2006)\textit{Wong, Wong, Pena, and
  Levasseur}}]{Wong2006}
Wong, C.~S., S.~K.~E. Wong, A.~Pena, and M.~Levasseur (2006), Climatic effect
  on DMS producers in the NE sub-Arctic Pacific: ENSO on the upper ocean,
  \textit{Tellus Series B-Chemical And Physical Meteorology}, \textit{58}(4),
  319--326.

\bibitem[{\textit{Yang et~al.}(1996)\textit{Yang, Zhang, Li, and
  Liu}}]{Yang1996}
Yang, G.~P., Z.~B. Zhang, L.~Li, and X.~Liu (1996), Study on the analysis and
  distribution of dimethyl sulfide in the East China Sea, \textit{Chinese
  Journal of Oceanology and Limnology}, \textit{14}(2), 141--147.

\bibitem[{\textit{Yang et~al.}(1999)\textit{Yang, Liu, Li, and
  Zhang}}]{Yang1999}
Yang, G.~P., X.~T. Liu, L.~Li, and Z.~B. Zhang (1999), Biogeochemistry of
  dimethylsulfide in the South China Sea, \textit{Journal Of Marine Research},
  \textit{57}(1), 189--211.

\bibitem[{\textit{Yvon et~al.}(1996)\textit{Yvon, Saltzman, Cooper, Bates, and
  Thompson}}]{Yvon1996}
Yvon, S.~A., E.~S. Saltzman, D.~J. Cooper, T.~S. Bates, and A.~M. Thompson
  (1996), Atmospheric sulfur cycling in the tropical Pacific marine boundary
  layer (12 degrees S,135 degrees W): A comparison of field data and model
  results. 1. Dimethylsulfide, \textit{Journal Of Geophysical
  Research-Atmospheres}, \textit{101}(D3), 6899--6909.

\end{thebibliography}
\end{document}
